Author's response to reviews

Title: Successful management of allergy to the insulin excipient metacresol in a child with Type 1 Diabetes: a case report

Authors:

Benjamin J Wheeler (ben.wheeler@otago.ac.nz)
Barry J Taylor (barry.taylor@otago.ac.nz)

Version: 2 Date: 11 March 2012

Author's response to reviews:

Thank you for your feedback. I have altered the manuscript as suggested. Sub headings have been removed from the case report and the consent section has been altered to reflect the dual consent that we have received.

Kind regards
Ben